Author's response to reviews

Title: Consumption of alcohol, cigarettes and illegal substances among physicians and medical students in Brandenburg and Saxony (Germany)

Authors:

Karen Voigt (Karen.Voigt@uniklinikum-dresden.de)
Sabine Twork (sabine_twork@hotmail.com)
Dirk Mittag (dirk.mittag@hausarzt-hlp.de)
Anne Göbel (la.pluie@web.de)
Roger Voigt (praxis.drvoigt@web.de)
Jörg Klewer (joklewer@aol.com)
Joachim Kugler (jokugler@aol.com)
Stefan R Bornstein (stefan.bornstein@uniklinikum-dresden.de)
Antje Bergmann (Antje.Bergmann@uniklinikum-dresden.de)

Version: 4 Date: 18 November 2009

Author's response to reviews: see over
Dear Sir or Madam,

thank you very much for sending the instructions for the final revision. You can find our replies below.

Major revisions

Figure legends: We provided a full description of the data within the legends.

Table 3: We cited the table within the main text.

Minor revisions

Placement text: We removed all figure and table placement text.

Tables: We removed all visible vertical lines from our tables.

Figures: We think that the figures are cropped as closely as possible.

Figure citations: We changed it and now we refer only “Figure” throughout the text.

We checked our manuscript for any typographical errors again and did any final corrections.

If there are any questions regarding the changed manuscript please contact me.

With kind regards

Karen Voigt (on behalf of the other authors)